CENWP-OP-B                                                                                                                         August 17, 2008
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 33 of 2008, which covers the period from 10 August through 16 August 2008.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 121.4 to 145.2 kcfs.  Daily average project tailwater ranged from 11.0’ to 12.6’ msl.  Daily average powerhouse forebay elevation ranged from 73.2’ to 75.6’ msl.  Daily average river temperature ranged from 69°F to 70°F.  Secchi disk measurements ranged from 4.5’ to 7.0’.
1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and B-branch, respectively, are included in the spill pattern.  Daily average spill ranged from 77.9 to 81.5 kcfs.
1.2.1 Bay 18 remains set at 2 dogs, or 2.9’, while the Bay 15 hoist is being repaired.
1.3 Unit Operation.  Powerhouse 2 is the priority powerhouse.
1.3.1 Unit 1 has remained out of service since 07 July for five year overhaul.  U1 was in operation for testing this week.  This testing resulted in a 1% violation at 1145 on 12 August.
1.3.2 Unit 2 went out of service at 0652 on 13 August to facilitate U1 efficiency testing.
1.3.3 Unit 7 has remained out of service since 04 June 2007 for turbine rehab.

1.3.4 Unit 9 has remained out of service since 04 August for turbine rehab.

1.3.5 Unit 11 has remained out of service since 0238 on 12 June due to rotor cracks.
1.3.6 Unit 15 was out of service from 0054 on 11 August until 1617 on 14 August for T12 maintenance.

1.3.7 Unit 16 was out of service from 0054 on 11 August until 1754 on 14 August for T12 maintenance.

1.3.8 Unit 17 was out of service from 0054 on 11 August until 1617 on 14 August for T12 maintenance.  U17 went out of service at 1611 on 15 August due to turbine bearing oil switch problems.

1.3.9 Unit 18 was out of service from 0054 on 11 August until 1836 on 14 August for T12 maintenance.
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby, as necessary, during nighttime hours to float debris.
1.4.1 Unit F2 was out of service from 0126 until 0214 on 15 August for switching.
1.5 Gatewell drawdowns are measured on Wednesdays.
2 ACTIVITIES

2.1 AWS Closures.  No AWS closures occurred this reporting week.
2.2 STS/VBS Inspections.  No STS/VBS video inspections occurred this week.  

Table 1.  STS and VBS inspections.  
	Unit
	Date
	Run Hours
	Comments

	11-18
	NA
	NA
	NA


2.3 Dewatering and Fish Salvage.  No dewatering or fish salvage activities occurred this reporting week.
3 RESEARCH  
3.1 Battelle.  BGS hydrophones are installed.  
3.2 Battelle.  Spillway survival evaluation continues.
3.3 CRITFC.  Annual chinook, coho, sockeye and steelhead sampling in the AFF continues.  Water temps reached 70°F on 08 August.  High temperature sampling protocols remain in effect.

3.4 CTUIR.  Received 192 lamprey for broodstock and 30 lamprey for radio-tagging from NOAA Fisheries as part of their Umatilla Basin restoration project, this season.
3.5 NOAA – Gilbreath.  Saltwater tag effects study continues.
3.6 NOAA – Jacobson.  Collected 25 subyearling chinook from PSMFC for pathogen testing.   
3.7 NOAA – Moser.  Trapping and PIT tagging at the AFF continues.  1375 lamprey have been trapped at the AFF this season.  Of the 1254 lamprey; 222 went to CTUIR, 20 to Warm Springs, 20 to UI for proximity analysis, 140 to UI for the flume testing, 286 were radio-tagged, 501 were PIT tagged and 285 were double tagged.  Water temps reached 70°F on 08 August.  High temperature sampling protocols remain in effect.
3.8 PSMFC Smolt Monitoring.  Condition, index, and GBT monitoring continue.  No adult salmonids were recorded crossing the separator bars.  25 subyearling chinook were collected for NOAA and 30 juvenile shad for USGS.  Water temps reached 70°F on 15 August.  High temperature sampling protocols went into effect.
3.9 University of Idaho.  Evaluating day, night, and LPS counts as well as detections from RT and HD PIT tag antennas to devise methods to estimate system wide passage of adult Pacific lamprey.  Have received 140 lamprey from NOAA Fisheries for lamprey flume work in the AFF.
3.10 USDA Wildlife Services. Avian hazing continues.
3.11 USGS. 30 juvenile shad were received for the Impacts of American Shad on the Columbia River study.  
3.12 WDFW Fish Counting.  Visual counts at Bradford Island and WA Shore continue.
3.13 Warm Springs.  Received 20 lamprey from NOAA Fisheries for ceremonial purposes this season.

4 FISHWAYS 

4.1 Project biologists inspected on 10-16 August.  
4.2  Fish Passage Plan Observations.
Table 2.  FPP Out of Criteria items.

	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND

	All week
	A branch gauge
	Reading <1.3
	Returned to non-shad mode early.
	See 4.5.4
	

	All week
	B branch gauge
	Reading <1.3
	Returned to non-shad mode early.
	See 4.5.4
	

	All week
	BI weir gauge
	Reading <1.2
	Returned to non-shad mode early.
	See 4.5.4
	

	8/10-11, 8/14-15
	FG3-27
	Incorrect position
	System in manual
	Ops adjusted
	

	CASCADES ISLAND

	8/10, 8/12-16
	UMT weir
	Reading  <1.3
	Returned to non-shad mode early.
	See 4.5.4
	

	8/10-11, 8/14
	FG6-14
	Incorrect position
	System in manual
	Ops adjusted
	

	WASHINGTON SHORE

	All week
	Weir 38
	Reading <1.3’
	Returned to non-shad mode early.
	See 4.5.4
	


4.2.1 FG6-10 closed.  The motor was repaired on 15 August 2007.  The shaft and gate have separated.  The diffuser will not be operational until repaired in winter 08/09.  It is OOC when TW is above 19.0’ msl.
4.2.2 PH1 elec. velocity meter is in an eddy.  Readings for FPP criteria are taken at the south end mechanical meter.

4.2.3 PH1 PCC diffusers FG2-22A and FG2-19 are closed or partially open.  Repairs scheduled for 09/10.  

4.2.4 FG3-5 closed.  The shaft and gate have separated.  Repairs scheduled for winter 09/10.  It is OOC when TW is above 16.7’ msl.

4.2.5 PH2 velocity meter.  The velocity meter is believed to be broken.  Repairs will not be made until winter 08/09.

4.3 Fishway Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore. Temperatures are in Fahrenheit.
4.3.1       Bradford Island temperatures are not available at this time
Table 3.  Bonneville Fishway Temperatures

	Date
	WS average
	AFF
	SMF

	8/10
	69.6
	69.6
	68.8

	8/11
	69.9
	69.9
	69.0

	8/12
	70.2
	70.3
	69.3

	8/13
	70.1
	70.2
	69.3

	8/14
	70.5
	70.6
	69.7

	8/15
	71.2
	71.3
	70.3

	8/16
	71.7
	71.7
	70.0


4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  

4.5 Adult Fishways. 
4.5.1 Calibration was checked on 15 August.
4.5.2 B-Branch and Cascades Island fishway PLC are in manual.
4.5.3 Removal of the FOG sea lion exclusion bars has not yet begun due to a broken lifting beam.  The four large orifices have bars (two at each end of the powerhouse). The stab plates remain installed in gates without exclusion bars.  
4.5.4 Shad passage mode concluded early, on 31 July, due to low numbers of shad passing BON.  Due to an oversight, this action had not been coordinated with FPOM and is thus recorded as out of criteria.
4.5.5 Cascades Island exit remains open to provide an exit for fish upstream of the UMT picket leads.
4.6 Juvenile Fishways.

4.6.1
   B2CC is scheduled to close on 02 September, as agreed to at the June FPOM meeting.

4.7 Lamprey Fishways.  
4.7.1 WS and BI AWS lamprey ramps remain in service.
4.7.2 The WS NDE lamprey ramp was shut off on 08 August due to high temperatures in the AFF and limitations on trapping at temperatures >70.0oF.  The trap will be operated four days a week until the daily average temperature in the AFF <69.5oF
5 CONSTRUCTION
5.1 PH1 ITS Wall Removal.  Three automated gates to be installed.  May slope the floor and remove the wall.
5.2 TIE crane.  The TIE crane boom repair is on hold until funding is secured.

6 HAZMAT, SPILLS AND CLEANUP.   Nothing to report this week.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A teletype sent on 19 June details the start of summer spill for juvenile fish passage.  It supercedes the teletype sent on 10 April. 

7.1.1 From 0001 on 21 July until 2400 on 31 August, Bonneville will spill 75K during daytime hours and to the 120% gas cap at night.

7.1.2 The spill pattern is found in section 2, Table BON-15 of the 2008 FPP, and daytime hours are defined in section 2, Table BON-5 of the 2008 FPP.  www.nwd-wc.usace.army.mil/tmt/documents/fpp/
7.1.3 At no time will the Project voluntarily spill above the 120% gas cap.  The 120% spill cap level for Bonneville can be found in the teletype “SPILL PRIORITY LIST”. 

7.1.4 At all times, maintain a minimum generation of 30kcfs and minimum spill of 50kcfs.

7.2 A teletype sent on 23 June authorizes the Project to dog open Spillbays 11, 14, and 18 at 2 stops (approx. 2.9’).  Spill shall be maintained as specified in 7.1.  This is to facilitate repairs to the Spillbay 15 hoist.  
7.3 A teletype sent on 04 August requests the Project to maintain Bonneville forebay elevation between 71.5 to 72.5 msl, hard constraint, 71.5 to 72.0 msl, soft constraint.  This operation is requested from 0800 to 1100 on 13 August.  This operation is requested to allow Spring Creek National Fish Hatchery to clean the lower portion of the fish ladders.

7.4 A teletype sent on 06 August authorizes the Project to deviate from unit priority from 06 August until 22 August to accommodate efficiency and relay testing of units 1 and 2. 
7.5 A teletype sent on 15 August requests the BON pool be operated with a 1.5’ range hard constraint and a 1.0’ range soft constraint from 0600 on 19 August until 1800 on 22 August.  This operation is intended to hold the pool as steady as possible to minimize debris in the tribal gill nets.
JAMES R. MAHAR, P.E.

Operations Manager
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